---
title: "Week 02"
subtitle: "SLR: Tidymodels + Model evaluation"
description: "Lab 01, Statistics experience assigned"
author: "Sep 05 - 09"
date: ""
tbl-colwidths: [60, 40]
title-block-style: none
---

::: callout-important
**Due dates:**

-   No AEs due this week
-   Lab 01:
    -   due Mon, Sep 12, 11:59pm (Thu labs)
    -   due Tue, Sep 13, 11:59pm (Fri labs)
:::

# Prepare

+-------------------------------------------------------------------------------------------------------------------------------------------------+-----------------------+
| Reading                                                                                                                                         | Corresponding lecture |
+=================================================================================================================================================+=======================+
| 📖 Introduction to Modern Statistics (IMS), Ch 7: [Linear regression with a single predictor](https://openintro-ims.netlify.app/model-slr.html) | Sep 05 & 07           |
+-------------------------------------------------------------------------------------------------------------------------------------------------+-----------------------+

# Participate

+--------------------------------------------------------------------------------+-------------+
| Topic                                                                          | Date        |
+================================================================================+=============+
| 💻 [SLR: Fitting models in R with tidymodels](../slides/03-slr-tidymodels.qmd) | Sep 05      |
+--------------------------------------------------------------------------------+-------------+
| 💻 [SLR: Prediction + model evaluation](../slides/04-slr-evaluation.qmd)       | Sep 07      |
+--------------------------------------------------------------------------------+-------------+
| 💻 [Lab 01](../slides/lab-01.qmd)                                              | Sep 08 & 09 |
+--------------------------------------------------------------------------------+-------------+

# Practice

+-----------------------------------------------------------------------+
| 📋 [AE 02 - Bike rentals in Washigton, DC](../ae/ae-02-bikeshare.qmd) |
+-----------------------------------------------------------------------+

# Perform

+---------------------------------+---------------------------------+
| Assignment                      | Due date                        |
+=================================+=================================+
| 📝 [Lab 01](../labs/lab-01.qmd) | Mon, Sep 12, 11:59pm (Thu labs) |
|                                 |                                 |
|                                 | Tue, Sep 13, 11:59pm (Fri labs) |
+---------------------------------+---------------------------------+
| 📝 Statistics experience        | Fri, Dec 09, 11:59pm            |
+---------------------------------+---------------------------------+
